**Author’s response to reviews**

**Title:** FANCM and RECQL genetic variants and breast cancer susceptibility: relevance to South Poland and West Ukraine

**Authors:**

Melissa Southey (msouthey@unimelb.edu.au)

Tu Nguyen-Dumont (tun@unimelb.edu.au)

Aleksander Myszka (amyszka@univ.rzeszow.pl)

Pawel Karpinski (polemiraza@poczta.fm)

Maria Sasiadek (maria.sasiadek@umed.wroc.pl)

Hayane Akopyan (akopyan.h@ihp.lviv.ua)

Fleur Hammet (fhammet@unimelb.edu.au)

Helen Tsimiklis (htso@unimelb.edu.au)

Daniel Park (djp@unimelb.edu.au)

Bernard Pope (bjpope@unimelb.edu.au)

Ryszard Slezak (slezak@gen.am.wroc.pl)

Nataliya Kitsera (nkitsera@gmail.com)

Aleksandra Siekierzynska (ola_sk@wp.pl)

**Version:** 2  **Date:** 04 Jan 2018

**Author’s response to reviews:**

The technical comments have been addressed in the attached files. Figure 1 has been uploaded. Declaration section is complete.